\documentclass[aps,prl,twocolumn,groupedaddress]{revtex4}

\usepackage{graphicx}
\usepackage{epstopdf}
\usepackage{natbib}
\usepackage{hyperref}

\newcommand{\ket}[1]{|{#1}\rangle}
\newcommand{\bra}[1]{\langle{#1}|}

\begin{document}
\begin{thebibliography}{28}
\expandafter\ifx\csname natexlab\endcsname\relax\def\natexlab#1{#1}\fi
\expandafter\ifx\csname bibnamefont\endcsname\relax
  \def\bibnamefont#1{#1}\fi
\expandafter\ifx\csname bibfnamefont\endcsname\relax
  \def\bibfnamefont#1{#1}\fi
\expandafter\ifx\csname citenamefont\endcsname\relax
  \def\citenamefont#1{#1}\fi
\expandafter\ifx\csname url\endcsname\relax
  \def\url#1{\texttt{#1}}\fi
\expandafter\ifx\csname urlprefix\endcsname\relax\def\urlprefix{URL }\fi
\providecommand{\bibinfo}[2]{#2}
\providecommand{\eprint}[2][]{\url{#2}}

\bibitem[{\citenamefont{Cirac et~al.}(1997)\citenamefont{Cirac, Zoller, Kimble,
  and Mabuchi}}]{cirac:1997}
\bibinfo{author}{\bibfnamefont{J.~I.} \bibnamefont{Cirac}},
  \bibinfo{author}{\bibfnamefont{P.}~\bibnamefont{Zoller}},
  \bibinfo{author}{\bibfnamefont{H.~J.} \bibnamefont{Kimble}},
  \bibnamefont{and} \bibinfo{author}{\bibfnamefont{H.}~\bibnamefont{Mabuchi}},
  \bibinfo{journal}{Phys. Rev. Lett.} \textbf{\bibinfo{volume}{78}},
  \bibinfo{pages}{3221} (\bibinfo{year}{1997}).

\bibitem[{\citenamefont{Kimble}(2008)}]{kimble:2008}
\bibinfo{author}{\bibfnamefont{H.~J.} \bibnamefont{Kimble}},
  \bibinfo{journal}{Nature} \textbf{\bibinfo{volume}{453}},
  \bibinfo{pages}{1023} (\bibinfo{year}{2008}).

\bibitem[{\citenamefont{Duan et~al.}(2001)\citenamefont{Duan, Lukin, Cirac, and
  Zoller}}]{DLCZ:2001}
\bibinfo{author}{\bibfnamefont{L.-M.} \bibnamefont{Duan}},
  \bibinfo{author}{\bibfnamefont{M.~D.} \bibnamefont{Lukin}},
  \bibinfo{author}{\bibfnamefont{J.~I.} \bibnamefont{Cirac}}, \bibnamefont{and}
  \bibinfo{author}{\bibfnamefont{P.}~\bibnamefont{Zoller}},
  \bibinfo{journal}{Nature} \textbf{\bibinfo{volume}{414}},
  \bibinfo{pages}{413} (\bibinfo{year}{2001}).

\bibitem[{\citenamefont{Reiserer et~al.}(2014)\citenamefont{Reiserer, Kalb,
  Rempe, and Ritter}}]{Reiserer:2014}
\bibinfo{author}{\bibfnamefont{A.}~\bibnamefont{Reiserer}},
  \bibinfo{author}{\bibfnamefont{N.}~\bibnamefont{Kalb}},
  \bibinfo{author}{\bibfnamefont{G.}~\bibnamefont{Rempe}}, \bibnamefont{and}
  \bibinfo{author}{\bibfnamefont{S.}~\bibnamefont{Ritter}},
  \bibinfo{journal}{Nature} \textbf{\bibinfo{volume}{508}},
  \bibinfo{pages}{237} (\bibinfo{year}{2014}).

\bibitem[{\citenamefont{Stobi\'{n}ska et~al.}(2009)\citenamefont{Stobi\'{n}ska,
  Alber, and Leuchs}}]{Stobinska:2009}
\bibinfo{author}{\bibfnamefont{M.}~\bibnamefont{Stobi\'{n}ska}},
  \bibinfo{author}{\bibfnamefont{G.}~\bibnamefont{Alber}}, \bibnamefont{and}
  \bibinfo{author}{\bibfnamefont{G.}~\bibnamefont{Leuchs}},
  \bibinfo{journal}{Europhys. Lett.} \textbf{\bibinfo{volume}{86}},
  \bibinfo{pages}{14007} (\bibinfo{year}{2009}).

\bibitem[{\citenamefont{Wang et~al.}(2011)\citenamefont{Wang,
  Min\'ar, Sheridan, and
  Scarani}}]{Yimin:2011}
\bibinfo{author}{\bibfnamefont{Y.}~\bibnamefont{Wang}},
  \bibinfo{author}{\bibfnamefont{J.}~\bibnamefont{Min\'ar}}, \bibinfo{author}{\bibfnamefont{L.}~\bibnamefont{Sheridan}},
  \bibnamefont{and} \bibinfo{author}{\bibfnamefont{V.}~\bibnamefont{Scarani}},
  \bibinfo{journal}{Phys. Rev. A} \textbf{\bibinfo{volume}{83}},
  \bibinfo{pages}{063842} (\bibinfo{year}{2011}).

\bibitem[{\citenamefont{Ritter et~al.}(2012)\citenamefont{Ritter, Nolleke,
  Hahn, Reiserer, Neuzner, Uphoff, Mucke, Figueroa, Bochmann, and
  Rempe}}]{Ritter:2012}
\bibinfo{author}{\bibfnamefont{S.}~\bibnamefont{Ritter}},
  \bibinfo{author}{\bibfnamefont{C.}~\bibnamefont{Nolleke}},
  \bibinfo{author}{\bibfnamefont{C.}~\bibnamefont{Hahn}},
  \bibinfo{author}{\bibfnamefont{A.}~\bibnamefont{Reiserer}},
  \bibinfo{author}{\bibfnamefont{A.}~\bibnamefont{Neuzner}},
  \bibinfo{author}{\bibfnamefont{M.}~\bibnamefont{Uphoff}},
  \bibinfo{author}{\bibfnamefont{M.}~\bibnamefont{Mucke}},
  \bibinfo{author}{\bibfnamefont{E.}~\bibnamefont{Figueroa}},
  \bibinfo{author}{\bibfnamefont{J.}~\bibnamefont{Bochmann}}, \bibnamefont{and}
  \bibinfo{author}{\bibfnamefont{G.}~\bibnamefont{Rempe}},
  \bibinfo{journal}{Nature} \textbf{\bibinfo{volume}{484}},
  \bibinfo{pages}{195} (\bibinfo{year}{2012}).

\bibitem[{\citenamefont{Lvovsky et~al.}(2009)\citenamefont{Lvovsky, Sanders,
  and Tittel}}]{Lvovsky:2009}
\bibinfo{author}{\bibfnamefont{A.~I.} \bibnamefont{Lvovsky}},
  \bibinfo{author}{\bibfnamefont{B.}~\bibnamefont{Sanders}}, \bibnamefont{and}
  \bibinfo{author}{\bibfnamefont{W.}~\bibnamefont{Tittel}},
  \bibinfo{journal}{Nat. Photonics} \textbf{\bibinfo{volume}{3}},
  \bibinfo{pages}{706} (\bibinfo{year}{2009}).

\bibitem[{\citenamefont{Zhang et~al.}(2012)\citenamefont{Zhang, Liu, Zhou,
  Chuu, Loy, and Du}}]{Zhang:2012}
\bibinfo{author}{\bibfnamefont{S.}~\bibnamefont{Zhang}},
  \bibinfo{author}{\bibfnamefont{C.}~\bibnamefont{Liu}},
  \bibinfo{author}{\bibfnamefont{S.}~\bibnamefont{Zhou}},
  \bibinfo{author}{\bibfnamefont{C.-S.} \bibnamefont{Chuu}},
  \bibinfo{author}{\bibfnamefont{M.~M.~T.} \bibnamefont{Loy}},
  \bibnamefont{and} \bibinfo{author}{\bibfnamefont{S.}~\bibnamefont{Du}},
  \bibinfo{journal}{Phys. Rev. Lett.} \textbf{\bibinfo{volume}{109}},
  \bibinfo{pages}{263601} (\bibinfo{year}{2012}).

\bibitem[{\citenamefont{Aljunid et~al.}(2013)\citenamefont{Aljunid,
  Maslennikov, Wang, Lan, Scarani, and Kurtsiefer}}]{Syed:2013}
\bibinfo{author}{\bibfnamefont{S.~A.} \bibnamefont{Aljunid}},
  \bibinfo{author}{\bibfnamefont{G.}~\bibnamefont{Maslennikov}},
  \bibinfo{author}{\bibfnamefont{Y.}~\bibnamefont{Wang}},
  \bibinfo{author}{\bibfnamefont{H.~L.} \bibnamefont{Dao}},
  \bibinfo{author}{\bibfnamefont{V.}~\bibnamefont{Scarani}}, \bibnamefont{and}
  \bibinfo{author}{\bibfnamefont{C.}~\bibnamefont{Kurtsiefer}},
  \bibinfo{journal}{Phys. Rev. Lett} \textbf{\bibinfo{volume}{111}},
  \bibinfo{pages}{103001} (\bibinfo{year}{2013}).

\bibitem[{\citenamefont{Volz et~al.}(2012)\citenamefont{Volz, Reinhard, Winger,
  Badolato, Hennessy, Hu, and Imamoglu}}]{Volz:2012}
\bibinfo{author}{\bibfnamefont{T.}~\bibnamefont{Volz}},
  \bibinfo{author}{\bibfnamefont{A.}~\bibnamefont{Reinhard}},
  \bibinfo{author}{\bibfnamefont{M.}~\bibnamefont{Winger}},
  \bibinfo{author}{\bibfnamefont{A.}~\bibnamefont{Badolato}},
  \bibinfo{author}{\bibfnamefont{K.~J.} \bibnamefont{Hennessy}},
  \bibinfo{author}{\bibfnamefont{E.~L.} \bibnamefont{Hu}}, \bibnamefont{and}
  \bibinfo{author}{\bibfnamefont{A.}~\bibnamefont{Imamoglu}},
  \bibinfo{journal}{Nat Photon} \textbf{\bibinfo{volume}{6}},
  \bibinfo{pages}{605} (\bibinfo{year}{2012}).

\bibitem[{\citenamefont{Johne and Fiore}(2011)}]{Johne:2011}
\bibinfo{author}{\bibfnamefont{R.}~\bibnamefont{Johne}} \bibnamefont{and}
  \bibinfo{author}{\bibfnamefont{A.}~\bibnamefont{Fiore}},
  \bibinfo{journal}{Phys. Rev. A} \textbf{\bibinfo{volume}{84}},
  \bibinfo{pages}{053850} (\bibinfo{year}{2011}).

\bibitem[{\citenamefont{Rezus et~al.}(2012)\citenamefont{Rezus, Walt, Lettow,
  Renn, Zumofen, G\"otzinger, and Sandoghdar}}]{Rezus:2012}
\bibinfo{author}{\bibfnamefont{Y.~L.~A.} \bibnamefont{Rezus}},
  \bibinfo{author}{\bibfnamefont{S.~G.} \bibnamefont{Walt}},
  \bibinfo{author}{\bibfnamefont{R.}~\bibnamefont{Lettow}},
  \bibinfo{author}{\bibfnamefont{A.}~\bibnamefont{Renn}},
  \bibinfo{author}{\bibfnamefont{G.}~\bibnamefont{Zumofen}},
  \bibinfo{author}{\bibfnamefont{S.}~\bibnamefont{G\"otzinger}},
  \bibnamefont{and}
  \bibinfo{author}{\bibfnamefont{V.}~\bibnamefont{Sandoghdar}},
  \bibinfo{journal}{Phys. Rev. Lett.} \textbf{\bibinfo{volume}{108}},
  \bibinfo{pages}{093601} (\bibinfo{year}{2012}).

\bibitem[{\citenamefont{Wenner et~al.}(2014)\citenamefont{Wenner, Yin, Chen,
  Barends, Chiaro, Jeffrey, Kelly, Megrant, Mutus, Neill et~al.}}]{Wenner:2014}
\bibinfo{author}{\bibfnamefont{J.}~\bibnamefont{Wenner}},
  \bibinfo{author}{\bibfnamefont{Y.}~\bibnamefont{Yin}},
  \bibinfo{author}{\bibfnamefont{Y.}~\bibnamefont{Chen}},
  \bibinfo{author}{\bibfnamefont{R.}~\bibnamefont{Barends}},
  \bibinfo{author}{\bibfnamefont{B.}~\bibnamefont{Chiaro}},
  \bibinfo{author}{\bibfnamefont{E.}~\bibnamefont{Jeffrey}},
  \bibinfo{author}{\bibfnamefont{J.}~\bibnamefont{Kelly}},
  \bibinfo{author}{\bibfnamefont{A.}~\bibnamefont{Megrant}},
  \bibinfo{author}{\bibfnamefont{J.~Y.} \bibnamefont{Mutus}},
  \bibinfo{author}{\bibfnamefont{C.}~\bibnamefont{Neill}},
  \bibnamefont{et~al.}, \bibinfo{journal}{Phys. Rev. Lett.}
  \textbf{\bibinfo{volume}{112}}, \bibinfo{pages}{210501}
  (\bibinfo{year}{2014}).

\bibitem[{\citenamefont{Kolchin et~al.}(2008)\citenamefont{Kolchin,
  Belthangady, Du, Yin, and Harris}}]{Kolchin:2008}
\bibinfo{author}{\bibfnamefont{P.}~\bibnamefont{Kolchin}},
  \bibinfo{author}{\bibfnamefont{C.}~\bibnamefont{Belthangady}},
  \bibinfo{author}{\bibfnamefont{S.}~\bibnamefont{Du}},
  \bibinfo{author}{\bibfnamefont{G.~Y.} \bibnamefont{Yin}}, \bibnamefont{and}
  \bibinfo{author}{\bibfnamefont{S.~E.} \bibnamefont{Harris}},
  \bibinfo{journal}{Phys. Rev. Lett.} \textbf{\bibinfo{volume}{101}},
  \bibinfo{pages}{103601} (\bibinfo{year}{2008}).

\bibitem[{\citenamefont{Gulati et~al.}(2014)\citenamefont{Gulati, Srivathsan,
  Chng, Cer\`{e}, Matsukevich, and Kurtsiefer}}]{Gulati:2014}
\bibinfo{author}{\bibfnamefont{G.~K.} \bibnamefont{Gulati}},
  \bibinfo{author}{\bibfnamefont{B.}~\bibnamefont{Srivathsan}},
  \bibinfo{author}{\bibfnamefont{B.}~\bibnamefont{Chng}},
  \bibinfo{author}{\bibfnamefont{A.}~\bibnamefont{Cer\`{e}}},
  \bibinfo{author}{\bibfnamefont{D.}~\bibnamefont{Matsukevich}},
  \bibnamefont{and}
  \bibinfo{author}{\bibfnamefont{C.}~\bibnamefont{Kurtsiefer}},
  \bibinfo{journal}{arXiv:1402.5800 [PRA (to be published)]}.

\bibitem[{\citenamefont{Bader et~al.}(2013)\citenamefont{Bader, Heugel,
  Chekhov, Sondermann, and Leuchs}}]{Bader:2013}
\bibinfo{author}{\bibfnamefont{M.}~\bibnamefont{Bader}},
  \bibinfo{author}{\bibfnamefont{S.}~\bibnamefont{Heugel}},
  \bibinfo{author}{\bibfnamefont{A.~L.} \bibnamefont{Chekhov}},
  \bibinfo{author}{\bibfnamefont{M.}~\bibnamefont{Sondermann}},
  \bibnamefont{and} \bibinfo{author}{\bibfnamefont{G.}~\bibnamefont{Leuchs}},
  \bibinfo{journal}{New Journal of Physics} \textbf{\bibinfo{volume}{15}},
  \bibinfo{pages}{123008} (\bibinfo{year}{2013}).

\bibitem[{\citenamefont{Franson}(1992)}]{Franson:1992cl}
\bibinfo{author}{\bibfnamefont{J.~D.} \bibnamefont{Franson}},
  \bibinfo{journal}{Phys. Rev. A} \textbf{\bibinfo{volume}{45}},
  \bibinfo{pages}{3126} (\bibinfo{year}{1992}).

\bibitem[{\citenamefont{Santori et~al.}(2002)\citenamefont{Santori, Fattal,
  Pelton, Solomon, and Yamamoto}}]{Santori:2002}
\bibinfo{author}{\bibfnamefont{C.}~\bibnamefont{Santori}},
  \bibinfo{author}{\bibfnamefont{D.}~\bibnamefont{Fattal}},
  \bibinfo{author}{\bibfnamefont{M.}~\bibnamefont{Pelton}},
  \bibinfo{author}{\bibfnamefont{G.~S.} \bibnamefont{Solomon}},
  \bibnamefont{and} \bibinfo{author}{\bibfnamefont{Y.}~\bibnamefont{Yamamoto}},
  \bibinfo{journal}{Phys. Rev. B} \textbf{\bibinfo{volume}{66}},
  \bibinfo{pages}{045308} (\bibinfo{year}{2002}).

\bibitem[{\citenamefont{Dousse et~al.}(2010)\citenamefont{Dousse, Suffczynski,
  Beveratos, Krebs, Lemaitre, Sagnes, Bloch, Voisin, and
  Senellart}}]{Dousse:2010}
\bibinfo{author}{\bibfnamefont{A.}~\bibnamefont{Dousse}},
  \bibinfo{author}{\bibfnamefont{J.}~\bibnamefont{Suffczynski}},
  \bibinfo{author}{\bibfnamefont{A.}~\bibnamefont{Beveratos}},
  \bibinfo{author}{\bibfnamefont{O.}~\bibnamefont{Krebs}},
  \bibinfo{author}{\bibfnamefont{A.}~\bibnamefont{Lemaitre}},
  \bibinfo{author}{\bibfnamefont{I.}~\bibnamefont{Sagnes}},
  \bibinfo{author}{\bibfnamefont{J.}~\bibnamefont{Bloch}},
  \bibinfo{author}{\bibfnamefont{P.}~\bibnamefont{Voisin}}, \bibnamefont{and}
  \bibinfo{author}{\bibfnamefont{P.}~\bibnamefont{Senellart}},
  \bibinfo{journal}{Nature} \textbf{\bibinfo{volume}{466}},
  \bibinfo{pages}{217} (\bibinfo{year}{2010}).

\bibitem[{\citenamefont{Akopian et~al.}(2006)\citenamefont{Akopian, Lindner,
  Poem, Berlatzky, Avron, Gershoni, Gerardot, and Petroff}}]{Akopian:2006}
\bibinfo{author}{\bibfnamefont{N.}~\bibnamefont{Akopian}},
  \bibinfo{author}{\bibfnamefont{N.~H.} \bibnamefont{Lindner}},
  \bibinfo{author}{\bibfnamefont{E.}~\bibnamefont{Poem}},
  \bibinfo{author}{\bibfnamefont{Y.}~\bibnamefont{Berlatzky}},
  \bibinfo{author}{\bibfnamefont{J.}~\bibnamefont{Avron}},
  \bibinfo{author}{\bibfnamefont{D.}~\bibnamefont{Gershoni}},
  \bibinfo{author}{\bibfnamefont{B.~D.} \bibnamefont{Gerardot}},
  \bibnamefont{and} \bibinfo{author}{\bibfnamefont{P.~M.}
  \bibnamefont{Petroff}}, \bibinfo{journal}{Phys. Rev. Lett.}
  \textbf{\bibinfo{volume}{96}}, \bibinfo{pages}{130501}
  (\bibinfo{year}{2006}).

\bibitem[{\citenamefont{Strekalov et~al.}(1995)\citenamefont{Strekalov,
  Sergienko, Klyshko, and Shih}}]{Strekalov:1995}
\bibinfo{author}{\bibfnamefont{D.~V.} \bibnamefont{Strekalov}},
  \bibinfo{author}{\bibfnamefont{A.~V.} \bibnamefont{Sergienko}},
  \bibinfo{author}{\bibfnamefont{D.~N.} \bibnamefont{Klyshko}},
  \bibnamefont{and} \bibinfo{author}{\bibfnamefont{Y.~H.} \bibnamefont{Shih}},
  \bibinfo{journal}{Phys. Rev. Lett.} \textbf{\bibinfo{volume}{74}},
  \bibinfo{pages}{3600} (\bibinfo{year}{1995}).

\bibitem[{\citenamefont{Jen}(2012)}]{Jen:2012}
\bibinfo{author}{\bibfnamefont{H.~H.} \bibnamefont{Jen}}, \bibinfo{journal}{J.
  Phys. B: At. Mol. Opt. Phys.} \textbf{\bibinfo{volume}{45}},
  \bibinfo{pages}{165504} (\bibinfo{year}{2012}).

\bibitem[{\citenamefont{Hodgson and Weber}(1997)}]{hodgson:1997}
\bibinfo{author}{\bibfnamefont{N.}~\bibnamefont{Hodgson}} \bibnamefont{and}
  \bibinfo{author}{\bibfnamefont{H.}~\bibnamefont{Weber}},
  \emph{\bibinfo{title}{Optical Resonators: Fundamentals, Advanced Concepts,
  and Applications}} (\bibinfo{publisher}{Springer Verlag},
  \bibinfo{year}{1997}), ISBN \bibinfo{isbn}{9783540761372}.

\bibitem[{\citenamefont{Agarwal and Gupta}(1994)}]{Agarwal:1994}
\bibinfo{author}{\bibfnamefont{G.~S.} \bibnamefont{Agarwal}} \bibnamefont{and}
  \bibinfo{author}{\bibfnamefont{S.~D.} \bibnamefont{Gupta}},
  \bibinfo{journal}{Phys. Rev. A} \textbf{\bibinfo{volume}{49}},
  \bibinfo{pages}{3954} (\bibinfo{year}{1994}).

\bibitem[{\citenamefont{Heugel et~al.}(2010)\citenamefont{Heugel, Villar,
  Sondermann, Peschel, and Leuchs}}]{Heugel:2010}
\bibinfo{author}{\bibfnamefont{S.}~\bibnamefont{Heugel}},
  \bibinfo{author}{\bibfnamefont{A.}~\bibnamefont{Villar}},
  \bibinfo{author}{\bibfnamefont{M.}~\bibnamefont{Sondermann}},
  \bibinfo{author}{\bibfnamefont{U.}~\bibnamefont{Peschel}}, \bibnamefont{and}
  \bibinfo{author}{\bibfnamefont{G.}~\bibnamefont{Leuchs}},
  \bibinfo{journal}{Laser Physics} \textbf{\bibinfo{volume}{20}},
  \bibinfo{pages}{100} (\bibinfo{year}{2010}).

\bibitem[{\citenamefont{Srivathsan et~al.}(2013)\citenamefont{Srivathsan,
  Gulati, Chng, Maslennikov, Matsukevich, and Kurtsiefer}}]{Srivathsan:2013}
\bibinfo{author}{\bibfnamefont{B.}~\bibnamefont{Srivathsan}},
  \bibinfo{author}{\bibfnamefont{G.~K.} \bibnamefont{Gulati}},
  \bibinfo{author}{\bibfnamefont{B.}~\bibnamefont{Chng}},
  \bibinfo{author}{\bibfnamefont{G.}~\bibnamefont{Maslennikov}},
  \bibinfo{author}{\bibfnamefont{D.}~\bibnamefont{Matsukevich}},
  \bibnamefont{and}
  \bibinfo{author}{\bibfnamefont{C.}~\bibnamefont{Kurtsiefer}},
  \bibinfo{journal}{Phys. Rev. Lett.} \textbf{\bibinfo{volume}{111}},
  \bibinfo{pages}{123602} (\bibinfo{year}{2013}).

\bibitem[{\citenamefont{Liu et~al.}(2014)\citenamefont{Liu,
  Sun, Zhao, Zhang, Loy, and Du}}]{liu:14}
\bibinfo{author}{\bibfnamefont{C.}~\bibnamefont{Liu}},
  \bibinfo{author}{\bibfnamefont{Y.} \bibnamefont{Sun}},
  \bibinfo{author}{\bibfnamefont{L.}~\bibnamefont{Zhao}},
  \bibinfo{author}{\bibfnamefont{S.}~\bibnamefont{Zhang}},
  \bibinfo{author}{\bibfnamefont{M.~M.~T.}~\bibnamefont{Loy}},
  \bibnamefont{and}
  \bibinfo{author}{\bibfnamefont{S.}~\bibnamefont{Du}},
  \bibinfo{journal}{Phys. Rev. Lett.} \textbf{\bibinfo{volume}{113}},
  \bibinfo{pages}{133601} (\bibinfo{year}{2014}).

\end{thebibliography}



%\preprint{}

\title{Reversing the temporal envelope of a heralded single photon using a cavity}

\author{Bharath Srivathsan}
\altaffiliation{Center for Quantum Technologies, 3 Science Drive 2, Singapore  117543}
\author{Gurpreet Kaur Gulati}
\altaffiliation{Center for Quantum Technologies, 3 Science Drive 2, Singapore  117543}
\author{Alessandro Cer\`{e}}
\altaffiliation{Center for Quantum Technologies, 3 Science Drive 2, Singapore  117543}
\author{Brenda Chng}
\altaffiliation{Center for Quantum Technologies, 3 Science Drive 2, Singapore  117543}
\author{Christian Kurtsiefer}
\altaffiliation{Center for Quantum Technologies and Department of Physics,
  National University of Singapore, 3 Science Drive 2, Singapore  117543}
\email[]{christian.kurtsiefer@gmail.com}
\date{\today}
\begin{abstract}
We demonstrate a way to prepare single photons with a temporal envelope that resembles the time reversal of photons from the spontaneous decay process. We use the photon pairs generated from a time-ordered cascade decay:
the detection of the first photon of the cascade is used as a herald
for the ground-state transition resonant second photon.
We show how the interaction of the heralding photon with an asymmetric Fabry-Perot cavity
reverses the temporal shape of its twin photon from a decaying to a rising exponential envelope.
This single photon is expected to be ideal for interacting with two level systems.

\end{abstract}

% insert suggested PACS numbers in braces on next line
\pacs{37.10.Gh, % Atom traps and guides 
42.50.Ct,       % Quantum description of interaction of light and matter;
                % related experiments 
32.90.+a        % Other topics in atomic properties and interactions of atoms;
                % with photons (restricted to new topics in section 32) 
}

\maketitle
Absorption of a single photon by a single atom or an ensemble of atoms is an interesting problem from
a fundamental point of view, and is also essential for many quantum information protocols~\cite{cirac:1997,kimble:2008,DLCZ:2001,Reiserer:2014}.
One of the requirements for an efficient absorption is that the temporal shape of the incident photon 
is the time reversal of the photon from the spontaneous decay process~\cite{Stobinska:2009,Yimin:2011}.
Temporally shaped light pulses have been utilized in many recent experiments to achieve efficient interactions between light and matter~\cite{Ritter:2012,Lvovsky:2009}.
In particular, the advantage of using a rising exponential shaped single photon for absorption in an atomic ensemble was demonstrated in~\cite{Zhang:2012}, and shaped multiphoton pulses for exciting a single atom was demonstrated in~\cite{Syed:2013}.
% Interaction of single photons with other two-level systems such as quantum
% dots~\cite{Volz:2012, Johne:2011}, single molecules~\cite{Rezus:2012} and
% super-conducting circuits~\cite{Wenner:2014} are also currently being
% researched and require a ``time reversed" envelope of the photons for
% efficient interaction.
This advantage also  applies to interacting single photons with other systems such as quantum dots~\cite{Volz:2012, Johne:2011}, single molecules~\cite{Rezus:2012} and superconducting circuits~\cite{Wenner:2014}.


Efficient preparation of single photons with narrow bandwidth and a rising
exponential envelope is not trivial.
One solution is the direct modulation of a heralded photon generated by an atomic medium~\cite{Kolchin:2008}.
This technique results in unavoidable losses due to filtering.
We have previously demonstrated a scheme to generate single photons with a rising exponential shape 
by heralding on photon pairs produced by cascade decay~\cite{Gulati:2014}
without filtering.
The drawback of this scheme is that the photon with the rising exponential envelope is
not resonant with an atomic ground state transition.

In this letter, we combine the asymmetric cavity design used by Bader et al.~\cite{Bader:2013}
with the well known temporal correlation properties of photon pairs~\cite{Franson:1992cl}
to invert the temporal envelope of the generated photon pairs:
with the proper heralding sequence we obtain a rising exponential single photon resonant with 
a ground state transition of $^{87}$Rb.
This concept is not limited to atoms, but can be equally applied to other physical system with a cascade level structure to obtain such photons~\cite{Santori:2002, Dousse:2010, Akopian:2006}. A related idea has been used in the past for non-local dispersion cancellation~\cite{Strekalov:1995}.


The photons emerging from an atomic cascade decay have a well defined time
order. The first photon of the cascade (signal) is generated before the photon
resonant with the ground state (idler). 
The resulting state can be described by a two photon wave function~\cite{Jen:2012} of the form
\begin{equation}
  \label{eq:wavepacket}
  \psi(t_{s},t_{i})=A\,e^{-\,(t_{i}-t_{s})/2\,\tau}\,\Theta(t_{i}-t_{s})\,,
\end{equation}
where $t_{s}, t_{i}$ are the detection times of the signal and idler
photons, and $\Theta$ is the Heaviside step function. In this notation, the
probability of observing a pair is proportional to $\left|\psi(t_{s},t_{i})\right|^2$. 
The exponential envelope and the decay time $\tau$ is
a consequence of the atomic evolution of the cascade decay.
If the detection of a signal photon is used as herald, the idler mode has a
single photon state with a exponentially decaying temporal envelope starting
at $t_i=t_s$.
Similarly, if the detection of an idler photon acts as a herald,
the signal photon has an exponentially rising temporal envelope.

An asymmetric cavity with the appropriate parameters transforms a light field with an exponentially rising envelope into one
with a decaying envelope~\cite{Bader:2013}.
The main point of this work is that
coupling the signal mode into a properly tuned asymmetric cavity 
before heralding results in a ``time reversed'' envelope for the idler photon.

\begin{figure}
  \begin{center}
    \includegraphics[width=\columnwidth]{figure1.eps}
   \caption{\label{fig:schematic}Schematic of the four-wave mixing experiment in
    collinear geometry. IF$_1$, IF$_2$: interference filters, used to combine pump beams
    and to separate the photons pairs. SMF: Single mode optical fibers. M$_1$,
    M$_2$: cavity mirrors.
    The incoming and outgoing mode
    of the cavity are separated by a polarising beam splitter (PBS) and a quarter wave plate
    ($\lambda/4$). 
    \mbox{D$_1$, D$_2$}: silicon avalanche photodiodes (APD).
    The inset shows the cascade level scheme for generation of
    photon pairs in $^{87}$Rb.}
  \end{center}
\end{figure}

The asymmetric cavity is formed by a partially
reflective mirror M$_1$, and a highly reflective mirror M$_2$, see Fig.~\ref{fig:schematic}.
The effect of the cavity on the signal mode is described as a frequency-dependent phase
factor~\cite{hodgson:1997,Agarwal:1994},
\begin{equation}\label{eq:transformation}
	C(\delta')\,=\,
  \frac{\sqrt{R_1}\,-\,
  \,\sqrt{R_2}\,e^{i\,\delta'/\Delta\nu_f}}{1 - 
  \,\sqrt{R_1\,R_2}\,e^{i\,\delta'/\Delta\nu_f}}\,,
\end{equation}
where $R_{1,2}$ are the reflectivities of M$_1$ and M$_2$, $\Delta\nu_f$ is the free
spectral range of the cavity, and $\delta'$ the detuning from the cavity resonance.
For $R_2=1$, the transformation of the incoming mode is lossless, i.e.,
$|C(\delta')|=1$.

The cavity transforms the two-photon wavefunction in Eq.~(\ref{eq:wavepacket})
into the two-photon wavefunction $\tilde{\psi}(t_{s}, t_{i})$:
\begin{equation}\label{eq:tr2}
  \tilde{\psi}(t_{s},t_{i})=\mathcal{F}_s^{-1}\left[
     C(\omega_s-\omega_s^0-\delta) \cdot 
     \mathcal{F}_s\left[\psi(t_s, t_i)\right]
   \right]\,,
\end{equation}
where $\mathcal{F}_s$ denotes a Fourier transform from $t_s$ to  $\omega_s$,
and $\delta$ is the detuning of the cavity resonance from the signal
photon center frequency $\omega^0_s/2\pi$.

If the ring-down time of the cavity matches the coherence time $\tau$ of the
photon pair in Eq.~(\ref{eq:wavepacket}),
the resulting wavefunction is:
\begin{eqnarray}\label{eq:wavepacket1}\nonumber
	\tilde{\psi}(t_{s},t_{i}) =
		\frac{A}{\sqrt{1+4\,\delta^2\tau^2}}\,
		[&2\,\delta\;\tau\,e^{-(t_i-t_{s})/2\,\tau}\,\Theta(t_{i}-t_{s})\,\\
		&+ e^{(t_i-t_{s})/2\,\tau}\,\Theta(-t_{i}+t_{s})]
\end{eqnarray}
with an exponentially rising and an exponentially
decaying component. Their
relative
weight can be controlled by the detuning
$\delta$, and for $\delta\,=\,0$, a time-reversed version of
Eq.~(\ref{eq:wavepacket}) is obtained:  
\begin{equation}
\label{eq:wavepacket2}
\tilde{\psi}(t_{s},t_{i})\,=\,A\,e^{(t_{i}-t_{s})/2\,\tau}\,\Theta(-t_{i}+t_{s}).
\end{equation}

Heralding on the detection of 
the modified signal photon results in an 
idler photon state with a rising exponential
envelope, ending at $t_i=t_{s}$.
The cavity thus effects a reversal of the
temporal envelope of the heralded idler photons from an exponential decay to a
rise. 

To experimentally investigate this method, we used the setup shown in
Fig.~\ref{fig:schematic}. We generate time-ordered photon pairs by
four-wave mixing in a cold ensemble of $^{87}$Rb atoms in a cascade level
scheme. Pump beams at 795\,nm and 762\,nm excite atoms from the $5S_{1/2},F=2$
ground level to the $5D_{3/2},F=3$ level via a two-photon transition. The 776\,nm
(signal) and 780\,nm (idler) photon pairs emerge from a cascade decay back to
the ground level and are coupled to single mode fibers.
All four modes are collinear and propagate in the same direction.
The coherence time $\tau$ of the 
photon pairs is determined by a time-resolved coincidence measurement between
the  detection of signal and idler photons to be 5.9\,ns. Details about the
photon pair source can be found in~\cite{Srivathsan:2013}
and~\cite{Gulati:2014}.

One of the modes of the photon pairs (signal in Fig.~\ref{fig:schematic}) is
coupled to the fundamental transverse mode of an asymmetric cavity, formed by
mirrors M$_1$, M$_2$ with  radii of curvature of 100\,mm and 200\,mm,
respectively. We characterize the cavity using a frequency stabilized laser of 
wavelength 776\,nm. The reflectivity of mirror M$_1$ is determined by 
direct measurement with a PIN photodiode to be $R_1=0.9410\pm0.0008$. 
Transmission through the mirror M$_2$ and absorption by the mirrors 
leads to losses in the cavity. The loss per round trip is 
determined from the transmission through and the reflection from the cavity 
and is included in the reflectivity of M$_2$, $R_2=0.998\pm0.001$.
The mirrors are mounted on a $55$\,mm long fused silica spacer, corresponding to a 
free spectral range $\Delta\nu_f=2.7$\,GHz. Therefore, an incident photon of Fourier 
bandwidth $1/(2\pi\tau)=27$\,MHz interacts effectively with only one
longitudinal mode of the cavity, ensuring that Eq.~(\ref{eq:transformation}) is an
adequate model. Temperature control of the spacer allows precise tuning and stabilization of the resonance frequency of the cavity. The light reflected off the cavity is separated from
the incident mode by using a polarising beam splitter (PBS) and a quarter waveplate
($\lambda/4$).

\begin{figure}
  \begin{center}
    \includegraphics[width=\columnwidth]{figure2.eps}
    \caption{\label{fig:plot1}
      Transformation of the temporal envelope of the heralded idler photon
      from exponential decay to rise when the cavity is in the signal mode.
      The y-axis shows the coincidence rate $G_{si}^{(2)}$ between the detectors D$_1$ and D$_2$ as a function of the detection time difference.
      The dashed lines represent $\left|\tilde{\psi}(t_{s},t_{i})\right|^2$,
      obtained from the model described by Eq.~(\ref{eq:wavepacket1}) for the
      indicated cavity detunings $\delta$,
      with amplitude $A$ as the only free parameter used to fit the experimental points.}
  \end{center}
\end{figure}

We infer the temporal shape of the heralded photons
from the time distribution of the coincidence rate $G_{si}^{(2)}$ between the APDs D$_1$ and D$_2$ (time resolution $<1$\,ns).
In Fig.~\ref{fig:plot1}
we show
$G_{si}^{(2)}$ for three different cavity-photon detunings.
When the cavity resonance is tuned
far away from from the signal photon frequency $\omega_s^0$,
in this case about $\delta/2\pi=120$\,MHz,
the temporal envelope remains nearly unchanged
from the exponential decay obtained without cavity. 
Off-resonant coupling of the incident signal photon to the cavity leads 
to the residual coincidences at times $t_i-t_s\,<\,0$.
At $\delta/2\pi=27$\,MHz, 
the time distribution becomes a symmetric exponential, 
and on resonance, $\delta/2\pi=0$,
we obtain a rising exponential shape.
For the three detunings 
the measurement agrees with the shape expected from Eq.~(\ref{eq:wavepacket1}):
the exponential time constants remain unchanged
and the new temporal shapes are determined by the phase shift across the cavity resonance via Eq.~(\ref{eq:transformation}).

From the time distribution of the coincidence counts
it is evident that
the situation is
symmetrical to what we presented in~\cite{Gulati:2014}:
by heralding on the signal photon
we now obtain an idler photon with a rising exponential temporal envelope.
% 
This result, though predicted by the theory, is particularly exciting:
the idler photon is resonant with a ground state transition
and the obtained temporal envelope is 
similar to
the time reversal of the one obtained by spontaneous decay.
The only deviation from the predicted shape occurs for a short
time interval after the detection of the herald ($t_i-t_s > 0$).
We attribute this deviation to an imperfect matching between the signal and cavity spatial modes.

\begin{figure}
  \begin{center}
    \includegraphics[width=\columnwidth]{figure3.eps}
    \caption{\label{fig:plot2} 
    Temporal envelope of the idler photon when the cavity is aligned and tuned to resonance with the idler mode.
    The dashed line represents
    $\left|\tilde{\psi}(t_{s},t_{i})\right|^2$
    obtained from Eq.~(\ref{eq:tr2}) by swapping $i$ and $s$.
    Also in this case, the amplitude is the only free parameters used in the fit.}
  \end{center}
\end{figure}

To confirm the predictive power of our model,
we repeated the same experiment swapping the roles of the signal and idler modes.
This corresponds to swapping the subscripts $s$ and $i$
in Eqs.~(\ref{eq:tr2}) and~(\ref{eq:wavepacket1}).
Figure~\ref{fig:plot2} shows the 
time resolved coincidence rate $G_{si}^{(2)}$ between the signal 
and modified idler photons
with the cavity tuned to resonance with the idler central frequency.
In this case the cavity transforms the exponentially rising temporal envelope
into a more complex shape.
Our model describes accurately this complex shape, as can be seen from the dashed line in Fig.~\ref{fig:plot2}.

Using the same setup,
we can infer the population of the cavity mode as a function of time, 
and observe its dependence on the envelope of the incident photon. 
\begin{figure}
\begin{center}
  \includegraphics[width=\columnwidth]{figure4.eps}
  \caption{\label{fig:excitation} Mean photon number in the cavity
  estimated using Eq.~(\ref{eq:n_mean}).
  On the left, the detection of an idler photon is used as herald and the cavity is in the signal mode.
  In this case we observe the interaction of an exponentially rising waveform with the cavity.
  On the right, the roles of signal and idler are swapped and the cavity interacts with an exponentially decaying incident photon.
   }
\end{center}
\end{figure}
We estimate the mean photon number $\langle n(t) \rangle$ in the cavity
using an algorithm similar to the one presented in~\cite{Bader:2013}.
We compare the time distributions of coincidence rates
$G_{fr}^{(2)}$ and $G_{or}^{(2)}$
when the cavity is tuned far-off resonance ($\delta/2\pi=200$\,MHz) and on resonance ($\delta/2\pi=0$\,MHz) 
with the incident photons, normalized against the total far-off resonance
coincidences. 
\begin{equation}\label{eq:n_mean}
  \langle n(t)\rangle\,=\,
  \frac{
  e^{-\eta t \Delta\nu_f}
  \int\limits_{-\infty}^{t} \left[G_{fr}^{(2)}(t') -  G_{or}^{(2)}(t') \right] e^{\eta t' \Delta\nu_f}\mathrm{d}t'
  }
  {
  \int \limits_{-\infty}^{\infty} G_{fr}^{(2)}(t') \mathrm{d}t'
  }.
\end{equation}
We estimated $\eta$, that includes the cavity losses per round trip and
transmission through M$_2$, to be $0.002\pm0.001$.

When the cavity is exposed to the idler mode,
a  heralded single photon with decaying exponential envelope 
interacts with the cavity:
the mean photon number in the cavity reaches a maximum of $0.44\pm0.01$.
On the other hand, when the cavity is aligned in the signal mode we have a
heralded single photon with an increasing exponential envelope interacting with
the cavity;
in this case, $\langle n(t)\rangle$ reaches a maximum of $0.76\pm0.01$.
% 
As expected, the photon with the rising exponential waveform interacts more efficiently with the cavity.
Following the analogy in~\cite{Heugel:2010}, we expect this result to be extended to the probability of absorption of a single photon by a single atom. 
In the case of interaction with a single atom, 
it will be necessary to also match the bandwidth of the transition.
We have already demonstrated how it is possible to control the bandwidth of the photon generated by the cascade process by adjusting the optical density of the atomic medium~\cite{Srivathsan:2013}.

In summary, we have demonstrated a method to transform a heralded single
photon with a decaying exponential temporal envelope to a rising exponential
envelope using a cavity. Using this method, we obtain single photons that
resemble the time-reversed versions of photons from  spontaneous decay
process resonant to the D2 line of $^{87}$Rb atoms. Single photon states of
this envelope and bandwidth would be useful for transferring information from
photons back into atoms. As this time reversal technique can be used with photon
pairs from other sources with time-ordered emission, as found e.g. in
molecules and quantum dots, it completes the toolbox necessary to interconnect
stationary qubits in a complex quantum information processing scenario.

We would like to thank Prof. Gerd Leuchs for useful discussions.   
We acknowledge the support of this work by the National Research Foundation
(partly under grant No. NRF-CRP12-2013-03) \& Ministry of Education in
Singapore.

{\it Note added in proof:} We became aware that during the review process, a
  similar work independently performed by Liu et al. has been published
  \cite{liu:14}.

%\bibliographystyle{apsrev}
%\bibliography{time_reversal}{}


\end{document}
